Author's response to reviews

Title: Why is it difficult to implement e-health initiatives? A qualitative study.

Authors:

Elizabeth Murray (elizabeth.murray@ucl.ac.uk)
Joanne Burns (burnsj2@lsbu.ac.uk)
Carl May (c.r.may@soton.ac.uk)
Tracy Finch (tracy.finch@newcastle.ac.uk)
Catherine O'Donnell (kate.o'donnell@gla.ac.uk)
Paul Wallace (p.wallace@ucl.ac.uk)
Frances Mair (frances.mair@gla.ac.uk)

Version: 6 Date: 27 December 2010

Author's response to reviews:

Thank you for sending these proofs and requests for amendments. I have done all that was requested, namely:

1. Ensured the abstract in the web submission system is identical to that in the main text;
2. Provided a functioning weblink for reference 5;
3. Removed the full stop from the title;
4. Added the country (UK) to the authors’ affiliations;
5. Checked that my (corresponding author) email is the same in the manuscript and the web submission system (it was);
6. removed all tracking;
7. provided full details for references 25 and 47;
8. removed the visible lines in the tables;
9. deleted the additional files containing tables;

I trust this is all now satisfactory

Best wishes
Elizabeth Murray